Author's response to reviews

Title: Ectopic cervical thymoma in a patient with Myasthenia gravis

Authors:

TiHei Wu (easyflyhigh@ndmctsgh.edu.tw)
Jong-Shiaw Jin (jsjin@ndmc1.ndmctsgh.edu.tw)
Tsai-Wang Huang (chi-wang@yahoo.com.tw)
Hung Chang (hung@ndmctsgh.edu.tw)
Shih-Chun Lee (chestsurgerytsgh@gmail.com)

Version: 5 Date: 5 July 2011

Author's response to reviews:

Dear Editor:
Thanks for spending time reading my article. I will try my best to answer these comments, and hopes you will be satisfied with them.

Formatting changes to be made:
-----------------------------

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

Ans: Yes, I have added the full affiliations of each authors.

Competing interests: Please include a `Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write 'The authors declare that they have no competing interests'.

Ans: Yes, I have added the `Competing interests' section.

Authors' contributions: Please place the Authors' contributions section towards the end of the manuscript text, directly preceding the Acknowledgements.

Ans: Yes, I have added the Authors' contributions section.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Line numbers: Please remove these.

Ans: Yes, I have removed the line numbers

Highlighting/tracking: Please remove all highlighting and tracking from the
manuscript.
Ans: Yes, I have remove all highlighting and tracking from the manuscript.

Running Title and Keywords: please remove these.
Ans: Yes, I have remove the Running Title and Keywords

Email address: please include email addresses for all the authors on the title page.
Ans: Yes, I have added the email addresses for all the authors

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Ans: Yes, I have check my manuscript for any typographical errors.

Sincerely,

TiHei Wu

Department of Surgery, Tri-Service General Hospital, National Defense Medical Center, 325, Sec 2, Cheng-kung Rd, Taipei 114, Taiwan, ROC.

E-mail address: easyflyhigh@ndmctsgh.edu.tw
Tel: +00 886 2 287923311-12698